Author’s response to reviews

Title: Factors Associated to Referral of Tuberculosis Suspects by Private Practitioners in Bali Province, Indonesia

Authors:

I Wayan Gede Artawan Eka Putra (gedeartawan@gmail.com)
Ni Wayan Arya Utami (arya_utami@yahoo.com)
I Ketut Suarjana (iks_jana@yahoo.com)
I Made Kerta Duana (madekerta2na@yahoo.com)
Cok Istri Darma Astiti (cokisbali@yahoo.co.id)
Ari Probandari (ariprobandari@yahoo.com)
Edine W Tiemersma (TiemersmaE@kncvtbc.nl)
Chatarina Umbul Wahyuni (chatrin03@yahoo.com)

Version: 2 Date: 24 April 2013

Author’s response to reviews: see over
The Editor,
BMC Health Services Research

We would be very grateful for your kind consideration of this manuscript entitled:

Factors Associated to Referral of Tuberculosis Suspects by Private Practitioners in Bali Province, Indonesia

This study reports the factors that associated to referral of tuberculosis (TB) suspects by private practitioners. I have revised the manuscript especially the abstract and adding the competing of interest. This study is important because it was give a practice recommendation to National TB control program (NTP) to increase public-private-mix partnership especially in TB case finding.

We would be very grateful for your kind consideration.

Yours Sincerely

dr. I Wayan Gede Artawan Eka Putra
Author for correspondence